Author’s response to reviews

Title: A case of allergic bronchopulmonary aspergillosis successfully treated with mepolizumab

Authors:
Takeshi Terashima (terasima@tdc.ac.jp)
Taro Shinozaki (t_shinozaki_mr@yahoo.co.jp)
Eri Iwami (okavango33et@gmail.com)
Takahiro Nakajima (tnakajima@tdc.ac.jp)
Tatsu Matsuzaki (tmatsuzaki@tdc.ac.jp)

Version: 4 Date: 14 Mar 2018

Author’s response to reviews:

March 14, 2018

Dr. Terumi Midoro-Horiuti
Associate Editor
BMC Pulmonary Medicine

Dear Dr. Midoro-Horiuti:

We wish to re-submit our manuscript, “A case of allergic bronchopulmonary aspergillosis successfully treated with mepolizumab.” We thank you and the reviewers for your thoughtful suggestions and insights, and we believe that our manuscript has benefitted greatly from this input.

We want to thank the reviewers again for their thoughtful comments. We hope that our manuscript is now suitable for publication in BMC Pulmonary Medicine.
Sincerely,

Takeshi Terashima, M.D., Ph.D.,

Department of Respiratory Medicine,

Tokyo Dental College Ichikawa General Hospital,

5-11-13, Sugano, Ichikawa, Chiba, 272-0824, Japan

TEL: 81-47-322-0151

FAX: 81-47-325-4456

E-mail: terasima@tdc.ac.jp

Responses to Editor’s Comments

1. In the Authors’ Contributions section, authors' names have been changed using their full initials.

2. In the Authors' contributions section, the statement that all authors have read and approved the final manuscript has been added.

3. We have removed the CARE checklist from the file inventory.

4. We have uploaded our manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.